



Rooshenas, L., Elliott, D., Wade, J., Jepson, M., Paramasivan, S., Strong, S., ... Donovan, J. L. (2016). Conveying Equipoise during Recruitment for Clinical Trials: Qualitative Synthesis of Clinicians' Practices across Six Randomised Controlled Trials. PloS Medicine, 13(10), [e1002147]. DOI: 10.1371/journal.pmed.1002147, 10.1371/journal.pmed.1002147.s001, 10.1371/journal.pmed.1002147.s002, 10.1371/journal.pmed.1002147.s003, 10.1371/journal.pmed.1002147.s004

Publisher's PDF, also known as Version of record

License (if available): CC BY

Link to published version (if available): 10.1371/journal.pmed.1002147 10.1371/journal.pmed.1002147.s001 10.1371/journal.pmed.1002147.s002 10.1371/journal.pmed.1002147.s003 10.1371/journal.pmed.1002147.s004

Link to publication record in Explore Bristol Research PDF-document

This is the final published version of the article (version of record). It first appeared online via Public Library of Science at http://journals.plos.org/plosmedicine/article?id=10.1371/journal.pmed.1002147. Please refer to any applicable terms of use of the publisher.

University of Bristol - Explore Bristol Research

General rights

This document is made available in accordance with publisher policies. Please cite only the published version using the reference above. Full terms of use are available: http://www.bristol.ac.uk/pure/about/ebr-terms.html



Citation: Rooshenas L, Elliott D, Wade J, Jepson M, Paramasivan S, Strong S, et al. (2016) Conveying Equipoise during Recruitment for Clinical Trials: Qualitative Synthesis of Clinicians' Practices across Six Randomised Controlled Trials. PLoS Med 13(10): e1002147. doi:10.1371/journal. pmed.1002147

Academic Editor: Sanjay Basu, Stanford University, UNITED STATES

Received: June 1, 2016

Accepted: September 7, 2016

Published: October 18, 2016

Copyright: © 2016 Rooshenas et al. This is an open access article distributed under the terms of the Creative Commons Attribution License, which permits unrestricted use, distribution, and reproduction in any medium, provided the original author and source are credited.

Data Availability Statement: The dataset (transcripts) informing this study are available on request, by contacting leila.rooshenas@bristol.ac. uk or carmel.conefrey@bristol.ac.uk. The data have not been uploaded to a public repository due to privacy concerns, but the authors will be able to consider specific requests on a case-by-case basis.

Funding: This investigation was supported by the Medical Research Council (MRC) (http://www.mrc. ac.uk/) Collaboration and Innovation for Difficult Trials in Invasive procedures (ConDuCT-II) Hub for **RESEARCH ARTICLE**

Conveying Equipoise during Recruitment for Clinical Trials: Qualitative Synthesis of Clinicians' Practices across Six Randomised Controlled Trials

Leila Rooshenas¹*, Daisy Elliott¹°, Julia Wade¹°, Marcus Jepson¹, Sangeetha Paramasivan¹, Sean Strong¹, Caroline Wilson¹, David Beard², Jane M. Blazeby¹, Alison Birtle³, Alison Halliday⁴, Chris A. Rogers⁵, Rob Stein⁶, Jenny L. Donovan^{1,7}, ACST-2 study group, By-Band-Sleeve study group, Chemorad study group, CSAW study group, Optima prelim study group, POUT study group¹

1 School of Social and Community Medicine, University of Bristol, Bristol, United Kingdom, 2 Nuffield Department of Orthopaedics, Rheumatology and Musculoskeletal Sciences, University of Oxford, Oxford, United Kingdom, 3 Rosemere Cancer Centre, Royal Preston Hospital, Preston, United Kingdom, 4 St George's, University of London, London, United Kingdom, 5 Clinical Trials and Evaluation Unit, Bristol Royal Infirmary, School of Clinical Sciences, University of Bristol, Bristol, Bristol, United Kingdom, 6 University College London Hospitals, London, United Kingdom, 7 National Institute for Health Research Collaboration for Leadership in Applied Health Research and Care West (NIHR CLAHRC West), University Hospitals Bristol NHS Trust, Bristol, United Kingdom

• These authors contributed equally to this work.

¶ Membership of the ACST-2 study group, By-Band-Sleeve study group, Chemorad study group, CSAW study group, Optima prelim study group, and POUT study group is provided in the Acknowledgments. * Leila.rooshenas@bristol.ac.uk

Abstract

Background

Randomised controlled trials (RCTs) are essential for evidence-based medicine and increasingly rely on front-line clinicians to recruit eligible patients. Clinicians' difficulties with negotiating equipoise is assumed to undermine recruitment, although these issues have not yet been empirically investigated in the context of observable events. We aimed to investigate how clinicians conveyed equipoise during RCT recruitment appointments across six RCTs, with a view to (i) identifying practices that supported or hindered equipoise communication and (ii) exploring how clinicians' reported intentions compared with their actual practices.

Methods and Findings

Six pragmatic UK-based RCTs were purposefully selected to include several clinical specialties (e.g., oncology, surgery) and types of treatment comparison. The RCTs were all based in secondary-care hospitals (n = 16) around the UK. Clinicians recruiting to the RCTs were interviewed (n = 23) to understand their individual sense of equipoise about the RCT treatments and their intentions for communicating equipoise to patients. Appointments in Trials Methodology Research (MR/K025643/1). The funding sources for the recruitment substudies that informed this synthesis are, listed by RCT (in alphabetical order): Acst-2—National Institute for Health Research (NIHR) (http://www. nihr.ac.uk/) Health Technology Assessment (HTA) (HTA 06/301/233) and the Bupa Foundation; By-Band-Sleeve—NIHR HTA (HTA 09/127/53); CSAW —Arthritis Research UK (http://www.

arthritisresearchuk.org/) (19707); Feasibility RCT of definitive chemoradiotherapy or chemotherapy and surgery for oesophageal squamous cell cancer— NIHR Research for Patient Benefit (RfPB) (PB-PG-0807-14131); POUT—Cancer Research UK (CRUK) (http://www.cancerresearchuk.org/)

(CRUK/11/027); Optima prelim: NIHR HTA (HTA 10/34/01). JW was funded by the NIHR HTA Programme (Projects No. HTA 96/20/06, HTA 96/ 20/99; ISRCTN20141297) during the period when this study was conducted. JLD was supported by the NIHR Collaboration for Leadership in Applied Health Research and Care (CLAHRC) West at University Hospitals Bristol NHS Foundation Trust and an NIHR Senior Investigator award. JMB is supported by an NIHR Senior Investigator award. This article presents independent research funded by NIHR, Arthritis Research UK, CRUK, and the MRC. The funders had no role in study design, data collection and analysis, decision to publish, or preparation of the manuscript.

Competing Interests: The authors have declared that no competing interests exist. The authors declare no support from any organisation for the submitted work beyond those funding bodies declared in the manuscript.

Abbreviations: QRI, Qualitative Research Integrated within Trials Recruitment Intervention; QuinteT, Qualitative Research Integrated within Trials; RCT, randomised controlled trial. which these clinicians presented the RCT to trial-eligible patients were audio-recorded (n = 105). The appointments were analysed using thematic and content analysis approaches to identify practices that supported or challenged equipoise communication. A sample of appointments was independently coded by three researchers to optimise reliability in reported findings. Clinicians and patients provided full written consent to be interviewed and have appointments audio-recorded.

Interviews revealed that clinicians' sense of equipoise varied: although all were uncertain about which trial treatment was optimal, they expressed different levels of uncertainty, ranging from complete ambivalence to clear beliefs that one treatment was superior. Irrespective of their personal views, all clinicians intended to set their personal biases aside to convey trial treatments neutrally to patients (in accordance with existing evidence). However, equipoise was omitted or compromised in 48/105 (46%) of the recorded appointments. Three commonly recurring practices compromised equipoise communication across the RCTs, irrespective of clinical context. First, equipoise was overridden by clinicians offering treatment recommendations when patients appeared unsure how to proceed or when they asked for the clinician's expert advice. Second, clinicians contradicted equipoise by presenting imbalanced descriptions of trial treatments that conflicted with scientific information stated in the RCT protocols. Third, equipoise was undermined by clinicians disclosing their personal opinions or predictions about trial outcomes, based on their intuition and experience. These broad practices were particularly demonstrated by clinicians who had indicated in interviews that they held less balanced views about trial treatments. A limitation of the study was that clinicians volunteering to take part in the research might have had a particular interest in improving their communication skills. However, the frequency of occurrence of equipoise issues across the RCTs suggests that the findings are likely to be reflective of clinical recruiters' practices more widely.

Conclusions

Communicating equipoise is a challenging process that is easily disrupted. Clinicians' personal views about trial treatments encroached on their ability to convey equipoise to patients. Clinicians should be encouraged to reflect on personal biases and be mindful of the common ways in which these can arise in their discussions with patients. Common pitfalls that recurred irrespective of RCT context indicate opportunities for specific training in communication skills that would be broadly applicable to a wide clinical audience.

Author Summary

Why Was This Study Done?

- Randomised controlled trials (RCTs) are a cornerstone of evidence-based medicine, but they need to recruit sufficient numbers of patients if they are to provide answers to the clinical questions they intend to address.
- Front-line clinicians are increasingly responsible for recruiting patients into RCTs, but there are questions around how they reconcile their own views about RCT treatments

with the requirement to convey "equipoise" to patients (the principle that the patient would be neither advantaged nor disadvantaged if they were to receive any of the trial treatments under investigation).

• Clinicians' attempts to communicate equipoise have not yet been investigated in the context of real doctor-patient interactions, and there are no evidence-based recommendations for how clinicians should convey equipoise to patients who are eligible for RCT participation.

What Did the Researchers Do and Find?

- We aimed to investigate how clinicians attempted to communicate equipoise to patients across six RCTs, with the objectives of (i) identifying broad practices that support or hinder equipoise communication and (ii) examining how these practices compare with clinicians' intentions.
- We interviewed clinicians (n = 23) to understand their intentions for communicating equipoise to patients, and audio-recorded their discussions ("appointments") with RCT-eligible patients (n = 105) to investigate their actual practices.
- Despite clinicians' best intentions, equipoise communication was omitted or undermined in almost half of the appointments (n = 48) through commonly recurring practices that were apparent across the RCTs, including (i) offering treatment recommendations, (ii) providing imbalanced descriptions of trial treatments, and (iii) disclosing personal predictions about RCT outcomes.

What Do These Findings Mean?

- Communicating equipoise to patients is a delicate process that can be overridden or undermined through commonly recurring practices that emerge irrespective of clinical context.
- There are opportunities to create generic guidance for clinicians based on these common challenges, with a view to facilitating the communication of equipoise to RCT-eligible patients.
- Our findings were limited by our intentional focus on RCTs that were anticipated to experience recruitment difficulties, and thus this issue needs further consideration in RCTs that are deemed less challenging to recruit patients into.

Introduction

Optimising recruitment to randomised controlled trials (RCTs) is a priority for clinicians, funding bodies, and healthcare decision-makers. Increasingly constrained resources reinforce the need for efficient generation of high-quality evidence to optimise patient care and inform service provision. Despite stringent review processes, at least half of funded RCTs fail to meet

recruitment targets [1], and recruitment difficulties remain the most frequent reason for premature trial closure [2]. Not only are these difficulties associated with considerable financial waste [3], but poor recruitment can delay progress or leave unanswered clinical questions in underpowered or incomplete studies.

Attempts to unravel RCT recruitment difficulties have focused on the pivotal role of frontline clinicians as recruiters. Clinicians are on the interface of research and practice, and are often responsible for presenting the option of RCT participation to eligible patients. Clinicians' perceptions of "equipoise" has repeatedly been cited as an assumed barrier to RCT recruitment [4,5]. Equipoise is broadly understood as a state of "substantial uncertainty" around treatment superiority [6], though there are discordant views around the degree and nature of uncertainty needed to justify randomisation [6]. Clinicians' personal experiences and awareness of anecdotal or non-RCT evidence can disrupt their sense of equipoise. This has underpinned commentaries on why recruiting to RCTs can be so challenging—particularly in disciplines that have a propensity for swift and clear-cut decision-making (e.g., surgery) [7]. Equipoise at an individual clinician level ("individual equipoise") can thus be hard to achieve-especially where ingrained clinical practices are under evaluation. The concept of "community equipoise" (or "clinical equipoise") refers to the principle that there is no consensus in the clinical community about which treatment is best. In theory, this framing of equipoise enables clinicians to align their instinct to act in the individual patient's interest with the utilitarian endeavour of benefiting future populations [8]. Although this viewpoint has been critiqued [9], clinical equipoise is widely accepted as the fundamental prerequisite that justifies an RCT. The pragmatic definition of equipoise adopted in this study framed the concept as a state where patients would be neither "advantaged nor disadvantaged" if they were to receive any of the trial treatments (as far as existing evidence would dictate) [10]. The recruiting clinician's role was to convey this notion to eligible patients.

Previous research suggests that patients' reasons for declining trial participation are often based on their interpretations of clinicians' treatment preferences [11–14]; this raises questions about how successfully equipoise is conveyed to patients in practice. Interview studies have shown that clinical recruiters are not necessarily in individual equipoise [12,15–17] but assume that their personal views can be set aside during encounters with trial-eligible patients [4,18]. Although there has been extensive theoretical commentary on these issues [5,19,20], equipoise communication has not yet been investigated in practice. There is therefore little empirical evidence to inform practical guidance for clinicians, despite recognition that discussing equipoise is a core component of their roles. We set out to investigate how clinicians conveyed equipoise during recruitment appointments in ongoing RCTs, with a view to identifying practices that supported or hindered equipoise communication. A secondary objective of the study was to examine how clinicians' practices compared with their reported intentions.

Methods

Ethical approval for this study was granted by six UK research ethics committees, with full details (e.g., dates, reference numbers) presented in <u>S1 Text</u>.

This was a qualitative study that adopted ethnographic approaches. Data were available from three sources: (i) audio-recorded appointments in which clinicians presented RCTs to eligible patients ("recruitment appointments"); (ii) interviews with recruiting clinicians in which perceptions of equipoise about trial treatments were explored, to enable comparison of reported intentions and actual practices; and (iii) documentary analysis of trial protocols containing the scientific evidence underlying the RCT, to aid interpretation of observed practices (e.g., assessing accuracy of information provision).

Context and Data Sources

Data from 16 publicly funded UK RCTs were available for inclusion, with an eventual sample of six selected for the study (see below). The sampling frame of 16 RCTs spanned a range of specialties (e.g., orthopaedics, vascular medicine, oncology). All were set in UK National Health Service–funded hospitals and had experienced (or were anticipated to experience) recruitment challenges. Data collection and analysis for the present study were conducted by the Qualitative Research Integrated within Trials (QuinteT) research team [21]. The lead of the QuinteT team (J. L. D.) was a collaborator or co-applicant on the RCTs, contributing an embedded methodological intervention that aimed to optimise recruitment: the QuinteT Recruitment Intervention (QRI) [22]. Members of the research team had no direct involvement in investigating the scientific questions the RCTs were addressing, but worked on the QRI. The QRI comprised an in-depth investigation of the RCT's recruitment processes and sources of difficulty (Phase 1), followed by implementation of tailored strategies to address these challenges (Phase 2). Given this study's focus on clinicians' naturalistic practices, this paper draws exclusively on data collected prior to any intervention (i.e., Phase 1).

More than half of the RCTs in the sampling frame included surgical interventions (10/16). This reflected the team's tendency to integrate the QRI into RCTs that were deemed "difficult" for recruitment by the RCTs' chief investigators and funding bodies. RCTs with surgical interventions are often thought to pose recruitment challenges due to cultural issues and assumed difficulties in negotiating equipoise [7].

The QRI involved audio-recording recruitment appointments and interviewing recruiters to understand recruitment processes. The methods employed for audio-recording appointments and conducting interviews were the same for each RCT, as summarised below.

Audio-Recording Appointments

Research nurses, local lead investigators, and central clinical trials units distributed information sheets about the recording process to recruiters via email/in person and obtained written consent for all of their recruitment appointments to be recorded (subject to patients' permission). Research nurses and recruiters identified patients who were eligible to participate in each RCT and distributed information sheets about the audio-recording process in person/via post. Patients' written consent was obtained in advance of the first appointment. Where prior contact was not possible, verbal consent was sought to record the first appointment; this recording was deleted if written consent was not subsequently provided. All appointments took place in one of 16 National Health Service secondary care hospitals around the UK. Audio-recordings were periodically transferred to the research team via encrypted data storage devices or secure file transfer.

Interview Processes

Recruiters in each RCT were invited to take part in a semi-structured interview to establish their individual perceptions of equipoise around the trial treatments. All recruiters in the smaller-scale RCTs were invited to participate in this study. In larger multi-centre RCTs (e.g., with 20 centres or more), a purposeful mix of recruiters from centres that were recruiting well and centres that were struggling to recruit were invited for interview. Research nurses, local investigator leads, and central clinical trials units identified recruiting clinicians, distributed information sheets via email/in person, and passed on consenting individuals' contact details to the interviewers (L. R., D. E., M. J., S. P., and C. W.). Although recruitment was also undertaken by research nurses in some centres, the present study focused exclusively on clinical recruiters in light of the study's aims.

A standard topic guide was used to ensure key issues were consistently covered in interviews across the RCTs. This topic guide was modified as interviews progressed in each RCT to include emerging recruitment issues that were specific to that trial. The broad topics of equipoise and views about trial treatments were consistently covered across all interviews. Interviews took place over the telephone or on hospital premises, involved just the researcher and clinician participant, and were audio-recorded with permission. Most interviews were conducted prior to receipt of the recruiter's audio-recorded appointments, though there were a few cases where recording of appointments had already begun at the time of interview. In these cases, the interviewers did not discuss any of their insights from listening to the appointments and recruiters had not received any feedback or training on their practices based on the recordings. All participants provided written informed consent. Field notes were used to record contextual details that might have influenced interview conduct and ideas for addition to topic guides.

Sampling Criteria for Cross-RCT Investigation

Sampling for the present investigation involved (i) selecting the RCTs and (ii) selecting the audio-recorded appointments. A purposeful sample of four (of 16) RCTs was initially selected. Sampling was directed by the intention to include a range of clinical specialties and types of treatment comparison (e.g., trials comparing two or more existing treatments, trials comparing current practice with a new technology). Although the sampling frame included more RCTs with surgical interventions, we intentionally included two non-surgical RCTs to increase variation. Sampling was also partially informed by practical issues such as the number of audio-recorded appointments collected. Two further RCTs were later added to the sample to confirm/disconfirm emerging findings. Details of the RCTs are shown in Table 1. The balance of surgical to non-surgical RCTs in the final sample was reflective of the sampling frame.

Audio-recorded appointments from the six RCTs were eligible for inclusion if they took place prior to any feedback to or training of the clinician (based on the recorded appointments), and if they were led by a clinician who had also been interviewed.

A total of 105 recruitment appointments met the study inclusion criteria and were analysed, coupled with 23 interviews with the clinicians who led these appointments. All clinicians were secondary care consultants (surgeons, oncologists, and neurologists). Interviews lasted between 21 min and 92 min. The selected data collated for this study were collected between October 2010 and December 2014. None of the 23 clinicians or 105 patients withdrew from the study (i.e., following provision of consent for the interviews or audio-recorded appointments).

The distribution of appointments per trial and clinician is shown in Table 1. In all, 83 of 105 appointments were with patients who ultimately declined trial participation (79%), 19 with patients who agreed to randomisation (18%), and three with patients where the outcome was not known (3%).

Analysis

Analysis was conducted after all the data had been collected from the individual RCTs and was led by the corresponding author (L. R.), who had no involvement in the majority of the RCTs included in the sample (5/6). As such, analysis was primarily conducted by an "independent" researcher, who had no prior knowledge of links between the interviews with clinicians and their associated audio-recorded appointments. Analysis proceeded in three stages. First, analysis of audio-recorded recruitment appointments focused on how equipoise was conveyed in practice. This occurred in conjunction with reviewing the RCT protocols to understand the nature of equipoise underpinning each trial. Second, interview transcripts were analysed once initial analysis of appointments was complete. This allowed themes emerging from the audio-recorded

RCT Number Used in This Study (ISRCTN Reference)	Clinical Specialty	Description of RCT Comparison Groups	Clinician Identifier	Number of Recorded Appointments
RCT1 [23] (ISRCTN00786323)	General	Group 1: Surgical procedure 1	R1	19
	surgery	Group 2: Surgical procedure 2	R2	10
			R3	16
			R4	8
RCT2 [24] (ISRCTN98387754)	Oncology	Group 1: Adjuvant treatment	R5	6
		Group 2: Surveillance (with treatment offered on	R6	2
		signs of cancer recurrence, if appropriate)	R7	2
RCT3 [25] (ISRCTN42400492)	Oncology	Group 1: Adjuvant treatment	R8	4
		Group 2: Prognostic-test-directed adjuvant	R9	7
		treatment	R8 R9 R10 R11 R12 R13	2
			R11	1
CT4 [26] (ISRCTN21144362)	Vascular	Group 1: Surgical treatment Group 2: Stenting	R12	4
	medicine		R13	1
			R14	1
			R15	3
RCT5 [27] (ISRCTN33864128)	Orthopaedics	Group 1: Arthroscopy with surgical manipulation	R16	3
		Group 2: Arthroscopy alone Group 3: Active monitoring with specialist reassessment	R17	1
RCT6 [28] (ISRCTN89052791)	Oncology/	Group 1: Neoadjuvant treatment and surgery	R18	5
	surgery	Group 1: Definitive non-surgical treatment	R19	4
			R20	2
			R21	1
			R22	1
			R23	2

Table 1. RCT details and associated number of appointments recorded (shown by participating clinician).

doi:10.1371/journal.pmed.1002147.t001

appointments to be generated as inductively as possible, without being influenced by knowledge of clinicians' views and perspectives. Third, findings from interviews and audio-recorded appointments were compared to examine how reported intentions corresponded with actual practice.

Recruitment appointments were listened to in full, and any discussion about the trial and/or trial treatments was transcribed verbatim if a full transcript of the appointment was not available. Full transcripts were available for four of the six RCTs, with two RCTs having "targeted transcription". Material that was not transcribed verbatim included discussion related to "rapport building", appointment scheduling, aspects of history taking not relevant to the diagnosis or trial, and technical descriptions of procedures that were often consistent across appointments for a particular recruiter. Transcripts were analysed using thematic techniques, guided by constant comparison methods derived from grounded theory methodology [29]. This involved line by line coding and organisation of codes into themes as analysis progressed. A coding frame was inductively developed, although there were a priori interests in examining how equipoise was communicated. The broad definition of equipoise provided by Djulbegovic et al. [10] informed coding. For example, any discussion/practices that supported the idea that the patient would be neither advantaged nor disadvantaged if they were to receive any/either of the trial treatments was coded as an example of conveying equipoise. Discussion/practices that did not support the above idea were coded as practices that challenged equipoise communication. Analysis was iterative in that previously coded transcripts were revisited in light of new codes/themes that emerged from later recordings. Analysis initially focused on RCT1, which had yielded the largest number of recorded appointments provided by the largest number of

recruiters. Appointments in three other RCTs were subsequently analysed to confirm, challenge, and develop themes from RCT1. This was followed by analysis of two further RCTs to ensure saturation had been achieved. Saturation was defined as the point at which analysis of appointments from three consecutive RCTs resulted in no new themes or lines of enquiry. There was scope to continue sampling appointments from the 16 RCTs, but saturation was achieved once 105 appointments had been analysed from six RCTs.

Data from audio-recorded appointments were summarised in a matrix displaying each appointment (rows) against major themes from the final coding framework (columns). Rows were grouped according to clinician and RCT to facilitate comparison at three levels: (i) across a single clinician's appointments to identify typical practices, (ii) between different clinicians of the same RCT to identify within-trial patterns and variations, and (iii) across trials to identify patterns and variations in practice across RCTs.

Interviews were transcribed in full and analysed using constant comparison methods, as described above. Data were organised and stored in NVivo (version 10, QSR International) to facilitate analysis. Sections of transcript that related to issues of uncertainty, equipoise, and clinicians' views on the RCT treatments were coded in greater detail. Descriptive summaries of each clinician's account from interview data were arranged alongside summaries of their observed practices to allow comparison.

A number of quality control processes were employed to ensure rigour in the reported findings. L. R. analysed all the data, with a subset of recordings from each trial double-coded by two independent members of the research team (D. E. and J. W.). Coding of individual transcripts was discussed in face-to-face meetings. Differences in interpretation were resolved through joint re-examination of original transcripts. Differences, where apparent, related to subtle misunderstanding of an RCT's context (an isolated example) and researchers' inclinations to occasionally code data in alignment with their particular research interests (which, on discussion, were generally deemed out of remit for this study). Any other differences related to individual researchers' tendencies to code data at different levels (resulting in more or less depth) and differences in terminology. In the latter case, discussion in meetings reassured that this did not reflect different semantics. Findings were presented in team meetings to ensure these resonated with researchers who originally worked on the QRI integrated into each RCT. There was a conscious effort to search for negative cases (examples that contradicted the broad themes emerging from the data). These are reported in the findings, where relevant.

Results

Given the complex and novel approach to data collection and analysis of this study, the findings have been presented in an order that most clearly conveys the key messages to emerge from the study. A summary of clinicians' individual perspectives on equipoise is presented first, to provide a framework for drawing comparisons between reported intentions and actual practices. The findings from the audio-recorded appointments—the focus of this study—are presented next. References are made to clinicians' interview accounts where relevant to demonstrate how reported intentions compared with actual practices.

Interview Findings: Clinicians' Individual Perceptions of Equipoise

Most clinicians expressed commitment to the RCT, acknowledging uncertainty amongst the clinical community about how patients should be treated (community equipoise). All clinicians acknowledged the need for evidence and indicated that they were individually "uncertain" about optimal treatment. However, interview accounts revealed a spectrum of uncertainty, ranging from complete ambivalence to strong beliefs about the superiority of a particular



Fig 1. Clinicians' individual perceptions of equipoise, based on interview accounts.

doi:10.1371/journal.pmed.1002147.g001

treatment. On the basis of the congruence of these findings with previous research [4], clinicians could be considered to fall on a continuum ranging from "clear equipoise" (equivocal views about trial treatments) to "not in equipoise" (treatment preferences and/or unbalanced views about trial treatments). All clinicians reported being uncertain or "in equipoise" on at least one occasion during the interview, but approximately half presented treatment preferences or polarised views at other stages of the interview. Clinicians were therefore divided into two groups: "more balanced" and "less balanced" (Fig 1). Those categorised as more balanced (n = 10) consistently indicated that they viewed the trial treatments in equilibrium (Box 1).

Box 1. Interview Extracts Comparing More Balanced and Less Balanced Clinicians' Views on RCT Treatments

More Balanced Group

R5: Um, so in other words here equipoise, as I say I think I do have equipoise here because I really do not know which is the best treatment for a patient. (R5, RCT2—more balanced group)

R1: If I thought I knew what the answer was then I probably wouldn't be interested in being in the trial. (R1, RCT1—more balanced group)

R9: I don't know what results this is going to be, I think there remains a degree of uncertainty. It keeps me awake at night. (R9, RCT3—more balanced group)

Less Balanced Group

R7: [My] gut reaction is probably that it [treatment x] does something and something needs to improve the outcome of these patients really, but my normal practice is not to give [treatment x] because I have to do things that have got an evidence base. (R7, RCT2—less balanced group)

R4: I think we'll see what other trials have demonstrated. I think we'll see that [treatment *x*] is better initially. (R4, RCT1—less balanced group)

R2: I'm struggling personally with what we're going to do in the trial with the patients who for whatever reason don't get on with [treatment y]. (R2, RCT1—less balanced group)

Less balanced clinicians (n = 13) expressed at least one clear preference or polarised opinion about treatment superiority during their interview—either in a general sense or in relation to particular sub-groups of eligible patients (Box 1). Evidence to support categorisation of each of the 13 less balanced clinicians as such is presented in S1 Data.

Regardless of their individual perceptions of equipoise, all clinicians indicated that their aim was to inform patients about the RCT without betraying their personal views. This reflected these clinicians' appreciation for community equipoise, which in turn reinforced prior work conducted in this field [4]. Although some anticipated that they might inadvertently influence patient choice, most believed they could achieve neutral information provision:

I can be unbiased during a presentation to them. [||] If you want me to be part of the randomised trial then I can't, I have to be completely non-biased. (R4, RCT1—less balanced group)

I'm probably biased towards a surgical treatment slightly, but I try not to express that. (R20, RCT6—less balanced group)

The less balanced versus more balanced categorisation provided a framework for comparing clinicians' individual perceptions of equipoise with their practices during recruitment appointments.

Audio-Recorded Appointment Findings: Articulating Equipoise

Equipoise issues were discussed in most of the observed appointments (83 of 105, 79%). Although the term "equipoise" was generally not used, the principle was described in a variety of ways. Equipoise was most often discussed in collective ("we") terms (79 of 83 [95%] appointments), thus indicating a tendency for clinicians to convey community (rather than individual) equipoise during patient encounters. This included references to clinical communities not knowing which treatment is best and presentations of equipoise as a global phenomenon ("nobody knows which treatment is best"). Individual equipoise ("I don't know which is best") was expressed in only 27 of 83 appointments (33%), though this was always in conjunction with statements of collective equipoise.

Equipoise was most commonly portrayed through statements indicating an absence of knowledge about which treatment was best (e.g., "we don't know what's best") or through explanations of what the RCT was seeking to achieve (e.g., "we're trying to find out"). Statements specifying uncertainty were also used, albeit less frequently (e.g., "we are uncertain"). Equipoise was also occasionally communicated by emphasising treatment equivalence, most often through reassurance that either/any trial treatment was equally appropriate for the patient. These statements were not mutually exclusive, with most appointments incorporating one or more of these statements (Table 2).

Audio-Recorded Appointment Findings: Difficulties Conveying Equipoise

Although equipoise was discussed in most appointments, over half of the clinicians (n = 16) encountered difficulties in consistently conveying equipoise throughout their discussions with patients. Most of these clinicians belonged to the less balanced group (n = 13), although there were three more balanced clinicians who also encountered difficulties.

Three recurring practices obstructed or obfuscated communication of equipoise. These practices varied in subtlety and were identified across multiple RCTs, irrespective of context. The patterns were: (i) overriding equipoise (e.g., by recommending a treatment), (ii) undermining equipoise (e.g., by communicating a preference), and (iii) omitting equipoise (e.g.,

Type of Statement	Examples		Appointments That Included at Least One Example of the Statement Type, <i>n</i> = 83	
		Number	Percent	
Intentions of RCT: specification of what the RCT is seeking to achieve	"We're trying to find the answer"; "we have a study that's looking at comparing [treatment x] and [treatment y] to see which is better"	64	77%	
Absence of knowledge: specification of not knowing	"We don't know what's best"; "I would love to know which is better"	61	74%	
Limitations or absence of evidence	"There is not enough evidence to say"	28	34%	
Uncertainty: statements that allude to not being sure or certain	"We do these studies when there is uncertainty"; "I can't say for sure which is best"	27	33%	
Emphasising patient suitability for both/all treatments	"My assessment of you is that you would be suitable for either treatment"	22	27%	

Table 2. Clinicians' descriptions of the concept of equipoise to patients during RCT recruitment appointments.

doi:10.1371/journal.pmed.1002147.t002

avoiding presenting or discussing equipoise). Each of these is presented in more detail below. Overall, 48 of the 105 appointments (46%) demonstrated one or more of these difficulties.

Overriding equipoise. Equipoise was overridden in nine appointments by clinicians clearly recommending a treatment option. Five clinicians from five RCTs recommended treatments in at least one of their appointments with trial-eligible patients. Most of these recruiters (n = 4) had revealed discomfort with one or more of the trial treatments in their interviews. This discomfort was apparent in appointments when patients appeared unsure or ambivalent towards treatment options, or had particular characteristics that challenged that particular clinician's sense of equipoise. For example, R4 (RCT1; Box 2) had expressed in the interview a preference for patients to receive particular trial treatments on the basis of their age, but also expressed the intention to refrain from conveying these biases to patients. Despite this, the recruiter went on to provide treatment recommendations to a number of patients, justifying their advice on the basis of these characteristics. Similarly, R8 (RCT3; Box 2) had intended to convey equipoise to the full range of eligible patients, yet reverted to recommending a particular treatment when faced with patients whose clinical characteristics placed them near the limits of the trial's eligibility criteria (for the patient in R8's extract in Box 2, the estimated level of benefit of treatment x placed the patient within the trial's eligibility criteria; this RCT's justification was built on the premise that there is uncertainty around the eligible patient group's potential to benefit from treatment *x*).

Undermining equipoise: imbalances in treatment presentation, and disclosure of opinion. Another pattern of communication that was problematic was the more subtle undermining of equipoise. This was apparent through imbalanced descriptions of trial treatments, and disclosure of clinicians' opinions and predictions about RCT findings. Fourteen clinicians (from all six RCTs) demonstrated at least one of these patterns. All bar one of these clinicians belonged to the less balanced group, based on the views they had expressed in interviews.

Equipoise could be undermined by implying that one trial treatment was more established than another. This was particularly problematic in RCTs evaluating treatments that were already used in routine practice. The treatments in these RCTs were particularly prone to clinical practice variations based on clinician preference. Some recruiters used loaded descriptors such as "gold standard" or "global standard" in relation to one trial treatment, without further explanation that the comparator may be just as good or better. These practices contrasted with other recruiters (from the same RCTs) who introduced the treatments in more neutral terms Box 2. Extracts Comparing Clinicians' Views/Intentions for Communicating Equipoise as Reported in Interviews with their Practices of Providing Treatment Recommendations in Audio-Recorded Recruitment Appointments

R4, RCT1—Less Balanced Group

Clinician's views/intentions expressed in interview:

R4: Hmm, I have to say, I mean being honest, with the more elderly patients with more co-morbidities, diabetes, I tend to go straight for [procedure *x*] on those patients (in routine practice). Hmm, just because they've got a limited life expectancy. In the younger patients I get with the less co-morbidities, I certainly have a preference for [procedure *y*].

R4: I think you present it fairly and then leave it up to the patient to make their own mind up. [||] I can't then guide them into which one I think they should have.

Communication in practice:

R4 [at outset of appointment]: Have you thought about what sort of [treatment] you want?

Patient: Yes. [Patient explains preference for treatment *x*]

R4: Ok. I'm going to talk to you about that, because of your age. I'm going to have a chat with you|| and the way you [explains characteristic]. I might try to steer you in a different direction I'm afraid. [||] I'll explain that a little bit more. Let me go through the options||. I mean, once you've heard all the options, I'm going to ask you the same question and we'll have a chat about it. [Recruiter explains treatment options]

R4: And that's why I was going to ask you what do you think you should have done. Because with your age and [characteristic], you would do well with [treatment y]. [||] My feeling would be for [treatment y], um—because you're otherwise relatively fit. The only real illness you have is [names illness]. I'm just wary of putting someone so young through [treatment x].

R8, RCT3—Less Balanced Group

Clinician's views/intentions expressed in interview:

R8: Then you start talking about the numbers and then at the end of the consultation you say, "Actually you can see the benefit from [treatment *x*] isn't great, so normally this is a decision which is made by the patient and the doctor together."

Communication in practice:

R8: [So I think] I mean if you think about it in a stepwise way, leaving the trial to one side, do you think that level of benefit is enough to warrant [treatment *x*]?

Patient: Yes.

R8: Yes [overlaps]. And I would say || certainly your doctors would agree with you, and most patients would agree with you. So I think, the initial question is, should we proceed with [treatment x] || the answer is "yes".

(e.g., by discussing practice variations or emphasising current assumptions that both/all trial treatments were equivalent). Within-trial variations in different recruiters' descriptions of trial treatments are illustrated in <u>Box 3</u>. One of these quotes is from a more balanced clinician considered to be an outlier (R5), in that he/she did not follow the same pattern observed

throughout this study. Whilst most of the more balanced clinicians did not demonstrate any of the difficulties in equipoise communication encountered by the less balanced clinicians, R5 was an exception. Despite being categorised as more balanced on the basis of the interview, this clinician went on to undermine equipoise in practice.

Box 3. Communication in Practice: Extracts from Audio-Recorded Appointments Illustrating Clinicians' Variable Descriptions of Treatments Available in Routine Practice

R7: People historically have either done very close follow-up, because they've said, well look there's no evidence to give [treatment *x*]. Or they've done [treatment *x*] because they said there's no evidence not to give [treatment *x*]. (R7, RCT2—more balanced group)

R5: Well [||] let me tell you what [||] I think the global standard of care is. I think that, um|| so nobody's saying we would never give you [treatment *x*]. But we would not normally give [treatment *x*] at this point. (R5, RCT2—more balanced group; outlier)

R20: We've been very keen to try and compare the two treatments—certainly [treatment *x*] has always been held up as the gold standard, which is fair enough. (R20, RCT6—less balanced group)

R23: The difficulty is that there [||] isn't a right and a wrong way of treating it. There are two very different treatments available for it and at the moment we don't really know whether one is better than another. (R23, RCT6—more balanced group)

Treatment superiority was also implicitly communicated in some clinicians' descriptions of the research question. For example, some clinicians from RCT1 and RCT6 presented the trial as an investigation of whether treatment *x* was "as good as" treatment *y* (R4 and R20; Box 4). Both treatments from these RCTs were widely used (albeit variably) in routine practice. These statements and hypotheses did not reflect the information and evidence presented in the RCT protocols. Furthermore, these loaded descriptions were in stark contrast to more neutral descriptions of the treatments/RCT rationale provided by other clinicians recruiting to the same RCTs (e.g., R1 and R21; Box 4).

Box 4: Communication in Practice: Extracts from Audio-Recorded Appointments Demonstrating Within-RCT Variation in Recruiters' Descriptions of Trial Treatments

R4: Part of this trial is to establish whether if the follow up for [treatment y] is good, is it as good as [treatment x]. (R4, RCT1—less balanced group)

R1: We don't know if one [treatment] is better than the other or if they are just alternatives. (R1, RCT1—more balanced group)

R20: Whether [treatment x] has got a tiny benefit over the [treatment y] we don't know, that's what we are trying to find out. (R20, RCT6—less balanced group)

R21: Now, there is evidence—or the—people think there is reasonable evidence that both treatments have similar outcomes but we don't know that for sure. There's not been a trial to show if one treatment is better at curing you than another treatment. (R21, RCT6—more balanced group)

Patients' responses provided insight into how recruiters' descriptions of trial treatments could be interpreted (Box 5). Use of terms such as "gold-standard" and statements about one treatment being potentially "as good as" another were not supported by information stated in the RCT protocols. These descriptors had a clear influence on patients' interpretations of treatment safety and likelihood of effectiveness (Box 5, extracts 1–3). Some terms could be considered "loaded" based on the clinical context in which they were mentioned. For example, some descriptions of tumour removal in RCT6 implied greater assurance of therapeutic benefit with one procedure over the other (Box 5, extract 4).

Box 5. Communication in Practice: Extracts from Audio-Recorded Appointments Illustrating Patients' Interpretations of Imbalanced Treatment Presentations

Extract 1: R18, RCT6—Less Balanced Group

R18: And there have been some very good results when we do [treatment *x*], but they've never ever been compared with what has been, over many years, the gold standard treatment [which is treatment *y*].

Patient [later]: Right. Well, if I went into the study and obviously was offered [treatment *y*], and it didn't work?

Extract 2: R22, RCT6—Less Balanced Group

R22: One is—it's a traditional treatment which is practised mostly around the world —or the world most commonly—and also more traditionally is (treatment *x*).

Patient partner [later]: This has all coming about because of the new technology [treatment *y*] or whatever. I mean, if you're talking thirty, forty years ago, automatically you'd have had [treatment *x*].

Patient partner [later]: But I'd just say $I \parallel$ do we want to try the new technology, or stick with the old technology?

Extract 3: R4, RCT1—Less Balanced Group

R4 [after describing definite beneficial effects of treatment *x*]: However, if you use [treatment *y*], well, potentially you can get the health benefits that you want, but it does require a lot of input from you.

Patient: Looking at what you told us, [treatment *x*] would be the best option for me, because it's a quick answer to my health problems.

Extract 4: R18, RCT6—Less Balanced Group

R18: The first is [treatment *x*] to remove the cancer to give us the chance of cure. **R18** [later]: [Treatment *y*] is designed to shrink down the tumour and try and kill it

off.

Patient relative: The thing that's worrying me if she has [treatment y] || if that doesn't take it away || but if she has [treatment x] then it's gone.

Equipoise was also undermined by clinicians predicting RCT findings. This was demonstrated by half of the clinicians observed, mostly in the less balanced group. Some clinicians tempered their predictions about trial outcomes with uncertainty and reiteration of the trial aims:

R7: That's because I don't know whether giving [treatment x] now is any better than just closely following you up. My gut reaction is that it may be, but I can't prove, until we finish off this study. (R7, RCT2—less balanced group)

There were, however, also examples of clinicians expressing their predictions and opinions without reiterating uncertainty:

R5: We would be reluctant to recommend [treatment *x*] in the absence of evidence because I think if there is a benefit from [treatment *x*] it will be small. (R5, RCT2—more balanced group; outlier)

Four of the RCTs included examples of clinicians not fully and accurately communicating equipoise as it was conveyed in the RCT protocols. Examples of practices that jeopardised equipoise communication included presenting non-evidence-based information with certainty and showing confidence in the likely outcome of the RCT. The extracts below compare two clinicians' approaches to introducing equipoise in an RCT that compared two surgical interventions with conservative therapy (RCT5). The first clinician (R16) articulates the three-way uncertainty as per the trial protocol, whereas the second (R17) implies a degree of certainty that one treatment (surgery) works. Equipoise as communicated by R17 thus appeared to be about "how", rather than "whether", surgery was effective:

R16: The trouble for us at the moment is we don't know whether to be offering people this operation. [||] And, if it is to work, we don't know quite how it works. (R16, RCT5—less balanced group)

R17: How that surgery works is not at all clear. What we're doing at the moment is participating in a trial. The idea is that that should hopefully give us some information as to which of those forms of treatment is producing the improvement in the people who do get better. (R17, RCT5—less balanced group)

Over-expressing confidence in a particular trial arm without reiterating uncertainty was particularly evident in RCTs that compared current practice with novel approaches. Comparison of clinicians' communication in practice (in appointments) with their expressed views/intentions (in interviews) revealed that tendencies to present opinions as facts were likely to be unwitting:

R10 [in interview]: So I say [to patients] because we don't know that [novel technology] is useful, the way the trial works is that half the people will just get|| [goes on to explain randomisation]. (R10, RCT3—less balanced group)

R10 [in appointment]: I have no concerns that it's a dodgy [technology] or anything. I think it'll give us a good result. (R10, RCT3—less balanced group)

Omitting equipoise. Discussion of equipoise was omitted in 22 appointments, from five of the six RCTs. Some of these omissions were associated with the issues discussed above (e.g., clinician-initiated treatment recommendations). The remainder were characterised by particular patterns in the ways clinicians structured or led the discussion: clinicians leading these

appointments directly elicited patients' preferences and accepted these without exploration, or geared the appointments towards patients selecting a treatment. These patterns were demonstrated by clinicians from both the less balanced (n = 3) and more balanced (n = 2) groups, and thus occurred irrespective of clinicians' individual perceptions of equipoise. Some of these clinicians had, however, expressed particular areas of discomfort in their interviews: namely, difficulties around transitioning from routine practice to recruitment appointments (Box 6, R18 and R2) and concerns about how to deal with patient preferences without appearing coercive (Box 6, R22).

Box 6. Interview Extracts Providing Insights into Why Clinicians Omit Equipoise

R18: If you think about it we're often expecting patients to get involved in decisionmaking, and in other situations outside of randomised studies we might say to the patient, "Look, there's treatment A or there's treatment B—which would you prefer?", and then we're expecting patients to decide. (R18, RCT6—less balanced group)

R2 [in reference to routine practice]: [When patients are] very clear about why they want to have an operation, or which operation they want to have, then I used to heave a sigh of relief because I thought, "Great, this patient is on board, they are clearly engaged." [||] So the problem of course now is that we have a situation whereby people are coming to us and we want them to say, "I was rather hoping you'd tell me the difference between the operations so that you can help me make my mind up," because then you can say exactly "Actually wonderful, [||] we don't know which is the best operation." [||] So of course we actually have to, in a sense, undo what we've done. (R2, RCT1—less balanced group)

R22: If you won't accept a patient's choice then you've probably || you're in the grounds of coercion. (R22, RCT6—less balanced group)

Discussion

This study investigated how clinicians communicated equipoise in RCT recruitment appointments. Presenting and sustaining equipoise was found to be a challenging process that was highly entwined with clinicians' own perceptions of equipoise. Clinicians varied in terms of their individual sense of equipoise, but were committed to the RCTs and reported that they intended to present treatments to patients in a neutral and balanced way. These expressed intentions did not always translate into practice—especially when clinicians harboured individual preferences or polarised views about treatments. Although equipoise was articulated in most appointments, observations across the RCTs revealed common practices that compromised equipoise communication. Clinicians with treatment preferences or less balanced views overrode or undermined equipoise through offering treatment recommendations, presenting imbalanced descriptions of trial treatments, or making clear their predictions about trial outcomes. These difficulties arose despite clinicians' intentions to conceal or suspend their personal inclinations or beliefs.

A key strength of this study was its analysis of real (rather than reported) practices and events. Issues of recollection bias and selective reporting were thus minimised. Paired interview and observational data allowed novel comparison of intended and actual practices, providing potential explanations for some of the observed practices. The study drew on a large dataset derived from a range of RCTs and clinical contexts, and focused on common practices that were identified across these settings. As such, findings were not limited to a particular RCT context and are likely to be applicable more widely. A qualitative approach was crucial for this particular research question given the absence of prior empirical research and the need for an inductive approach. We made a conscious effort to look across a broad range of RCTs and focus on the practices that were common across these contexts. The reported findings are thus likely to be salient to clinicians across specialties.

The study had a number of limitations. Data were derived from RCTs that were experiencing (or were anticipated to encounter) recruitment difficulties. There was a prior intention to focus on challenging RCTs with a view to informing strategies to overcome recruitment difficulties, but our findings need to be considered further in RCTs that recruit well. Another limitation was the reliance on RCTs involving surgical interventions, which could have raised particular issues. Nonetheless, our sample also included RCTs that did not include surgical interventions, and although these were fewer in number, we did not identify any notable differences between RCT types in relation to the key findings. Integration of the recruitment substudy (QRI) across multiple RCTs meant that some processes (e.g., study invite distribution) were organised centrally (e.g., by clinical trials units) or influenced by local issues (e.g., research nurses' capacity and work patterns in RCT centres). We did not have a sense of how many patients and clinicians declined participation in our study (or were not approached), and as a consequence it is possible that the data captured provide insights from particular groups of patients/clinicians. We believe this is unlikely in the case of patients, as informal discussion with research nurses revealed that it was rare for patients to decline consent for audio-recording their appointments. If recordings were not captured, this tended to be due to logistical or organisation issues. However, findings may be limited by the self-selected clinicians who agreed to be interviewed and to have their appointments audio-recorded. As these individuals may have been particularly interested in improving communication or supportive of the RCT, it might be that problematic presentations of equipoise are more pronounced in appointments led by clinicians who might not have been as engaged with the research. The audio-recording process itself may have altered clinicians' practices, though, again, any Hawthorne effect is likely to have manifested as recruiters being more mindful of concealing their views. Some of the clinicians participating in the study contributed only one appointment for observation, which limited the scope to compare clinicians' reported and actual practices. The limitations relating to the lack of information about clinicians/patients who did not volunteer/agree to participate, and the potential influence of the Hawthorne effect, are partly countered by the fact that we identified similar challenges across clinicians/RCTs. This lends some assurance to the integrity of the key findings presented. Finally, like any interview-based study, it is possible that participants' accounts were influenced by their knowledge of the researchers' interests (in this case, recruitment). Clinicians were nonetheless not aware of our specific interest in investigating equipoise.

Much of the literature around clinical equipoise has been theoretical, although there have been some empirical studies that have focused on clinicians' reported difficulties with negotiating equipoise. These studies indicate that clinicians often assume that personal struggles with equipoise can be set aside during patient encounters and have no bearing on recruitment [4,18]. However, studies reporting patient perspectives indicate that their decisions around trial participation can be guided by their interpretations of what their clinician prefers or recommends [11–14]. There is also evidence to suggest that patients construct these interpretations based on perceived cues, rather than direct statements [14]. The present study bridges these bodies of evidence by providing insight into what actually happened during particular RCT recruitment appointments. This has enabled a detailed examination of how equipoise is put into practice, and identification of commonly recurring ways in which equipoise communication can be compromised.

Some of our findings are supported by previous research involving observations of recruitment appointments. These studies were largely nested in single RCTs and reported on trialspecific challenges clinicians encountered. For example, the potential for loaded terminology to disrupt equipoise has been reported in RCTs of laryngeal and prostate cancer [30,31]. The scope to look across RCTs helped us to distinguish particular aspects of discussions that are prone to loaded terminology or misrepresentation, such as portrayal of current practice and descriptions of the RCT question [30]. These areas could be amenable to generic training or guidance that encourages clinicians to carefully articulate the information specified in the trial protocol.

Though there are few multi-RCT studies of communication practices, one study by Jenkins and colleagues included a content analysis of how frequently particular trial concepts were mentioned in recruitment appointments [32]. Although "equipoise" was not specifically considered, the authors reported that "uncertainty" was mentioned in almost all of the 82 appointments observed. By contrast, we found equipoise discussion was omitted in 20% of the observed appointments. This discrepancy may be explained by differences in sample sizes and clinical contexts. Jenkins and colleagues observed fewer clinicians' practices (n = 5) and focused solely on oncology trials. Oncologists are more accustomed to discussing issues of equipoise and uncertainty, compared to clinicians working in other specialties, such as surgery [7]. The greater diversity of clinical specialties and greater number of clinicians observed (n = 23) in our study may explain the comparatively higher proportion of recruitment appointments that omitted discussion of equipoise/uncertainty. Other studies have also shown that equipoise is often omitted from RCT recruitment discussions, though these studies also specifically measured the frequency of statements of "uncertainty" [33,34]. Reported presentations of equipoise in this study were based on clinicians' own terms, rather than preconceived assumptions. This confirmed that statements of equipoise often extend beyond mere mention of uncertainty, as suggested in prior descriptions of how clinicians discuss equipoise in interviews [35].

Introducing the concept of equipoise to patients can be a challenge in its own right, as indicated by our findings that some clinicians were inclined to omit the concept from recruitment appointments. The substantial literature on clinicians' management of medical uncertainty may be particularly helpful for illuminating explanations for the omission of equipoise. Uncertainty is a complex, multifaceted phenomenon that is pervasive throughout the medical literature. The definition of equipoise used in this study most closely aligns with the notion of scientific ambiguity—uncertainty in clinical recommendations on the basis of insufficient scientific evidence [36]. There has been remarkably little research that investigates how scientific ambiguity is operationalised in routine practice, although the small body of evidence that does exist has highlighted two broad strategies adopted by clinicians: creating an illusion of certainty [37–39] and transferring uncertainty to others (such as the patient) [37,38,40,41]. Our observations of clinician-initiated recommendations and tendencies to gear appointments towards patients choosing a treatment may therefore reflect strategies used to address uncertainty in routine practice, and underlie why recruitment needs to be recognised as a different activity [42]. There are opportunities for future research to compare recruitment appointments with routine consultations to explore issues arising from these different contexts.

The concept of equipoise is poorly demarcated in the literature, and, as such, the findings of this study need to be interpreted in light of the assumptions we have made and the stances we have taken. Equipoise can be considered in two contexts: first, as a phenomenon that justifies an RCT proposal and, second, as a rationale for randomisation presented during an RCT

recruitment appointment. This study assumed that public funding of the RCTs indicated there was sufficient justification for the RCTs, although questions have also been raised around whether equipoise should take account of considerations such as social values and cost of treatment [43]. Definitions of equipoise in the context of the RCT recruitment appointment can also vary. Rather than an inductive grounded theory investigation [29], we specified a priori a broad and pragmatic definition of equipoise as a state where a patient would be neither advantaged nor disadvantaged if they were to receive any of the RCT treatments under investigation, based on existing evidence. We assumed that a patient who is clinically eligible for an approved RCT should be equipped with evidence-based information that ensures they are aware that the interventions under consideration would be safe, equivalent, and appropriate for them. This does not, however, account for external or patient-related non-clinical factors that can alter this balance. Future research needs to examine how patients' values, beliefs, and preferences are managed in the RCT recruitment appointment, and the implications this has for recruitment outcomes. Focusing on recruiters' explanations and information provision in isolation risks creating an oversimplified picture of what is likely to be a complex set of interactions. Our findings may also have oversimplified some issues through our pragmatic labelling of clinicians as more or less balanced. This labelling acted as a proxy for individual equipoise but does not take account of the continuum of views we identified. Our findings should thus be viewed as an initial step, in need of future refinement.

The findings of this study have practical implications. Identification of commonly recurring communication challenges provides opportunities for targeted strategies to improve practice. The practices that obfuscated equipoise communication varied in nature and would require different solutions. Treatment recommendations were the most extreme practices that contradicted equipoise. Recommendations were often rationalised in terms of patients' characteristics, presented by clinicians who had alluded to being uncomfortable with particular aspects of trial eligibility criteria. Previous research has also shown that clinicians' perceptions of equipoise can waver at the limits of eligibility criteria [4]. Like any aspect of clinical practice, patients' suitability for trial interventions needs to take into account individual patient circumstances. However, difficulties arise when patients are steered against RCT participation on the basis of clinical or biological characteristics that are not stated as formal exclusion criteria. Filtering patients on these bases can have implications for the generalisability of RCT findings and interfere with eligible patients' autonomy to make decisions about trial participation. One solution to untangling these difficulties might be to promote open discussion about discomfort or concerns about trial eligibility criteria during trial initiation meetings and site visits [4,42]. The QRI uses evidence from audio-recorded appointments and interviews to air these issues during investigator feedback and training sessions [22]. Recording detailed reasons for patients not entering a trial (e.g., in screening logs) will also lend some transparency to trial conduct. These reported details will need to go beyond ambiguous statements such as "clinician discretion". Another alternative is for nurses or other professionals to undertake recruitment. While nurses do increasingly take on this responsibility, clinicians need to be accustomed to presenting equipoise and uncertainty comfortably: patients will still look to clinicians for advice, irrespective of whether clinicians are actively recruiting into an RCT. Furthermore, there is encouraging emerging evidence that it is possible for clinicians to learn recruitment skills [22,44,45].

This study found that clinicians often shared their opinions and predictions about RCT outcomes with patients. This raises questions around what should be expected of clinicians who hold the dual roles of physician and RCT recruiter. Avoiding personal views may be difficult in situations where patients request an opinion or recommendation. Trials that involve treatments that differ to usual care pose particular challenges, as clinicians need to have a degree of faith in "the new" to justify disrupting usual practice. Conversely, clinicians may also need to justify use of older, less innovative treatments in trials that evaluate novel interventions that have already diffused into routine practice without a supporting evidence base. Consistently communicating the concept of community equipoise throughout the appointment is one potential solution to responding to recommendation requests. The principles underpinning Chard and Lilford's distinction between absolute equipoise and effective equipoise may also be useful here, in which equipoise around a single outcome (absolute equipoise) is distinguished from equipoise created by balancing multiple considerations (e.g., financial costs, side effects, effectiveness, quality of life) (effective equipoise) [5]. This alternative approach to viewing equipoise may provide a helpful framework in practice for balancing the advantages and disadvantages of trial treatments in efforts to sustain equipoise.

The common occurrence of particular equipoise-related difficulties across trials suggests that these challenges are amenable to generic training that could be incorporated into undergraduate and postgraduate medical curricula—particularly given that clinicians are increasingly expected to discuss issues of uncertainty and evidence. Recommendations for improving communication may also be adopted if clinicians are able to consider and apply these in the context of an RCT they are working within. A current programme of applied research has been developed to address recruitment difficulties by delivering tailored interventions as RCTs proceed [22]. Interventions applied within RCTs to date have included training around some of the issues reported in the present study [45]. Our findings also raise the possibility of targeting training to recruiters based on their individual perceptions of equipoise expressed at the trial's outset (e.g., through interviews). Further research will need to investigate whether training can lead to sustained change in the practices of less balanced clinicians.

This study did not intend to make causal inferences between recruiters' practices and trial participation outcomes, though there is potential for future research to examine possible associations. There is also potential to consider how categorisation of recruiters as more or less balanced is associated with recruitment outcomes, although this would run counter to our interpretation of individual equipoise falling on a continuum. Such efforts will need to consider the complex nature of patient decision-making, which can be influenced by an array of factors within and outside the recruitment appointment. Future research may also consider patients' interpretations of appointment events—particularly their perspectives on equipoise and trial participation immediately following appointments, and levels of informed consent.

Conclusion

Communicating equipoise in practice is a delicate process that can be challenging, especially when clinicians are negotiating their own struggles with equipoise. Despite clinicians' assumptions that personal biases and preferences can be set aside, these can unwittingly materialise in encounters with patients. This phenomenon calls for careful reflection and training to enable clinicians to communicate equipoise as they intend. Guidelines may need to consider the possibility of offering training based on clinicians' conveyed (rather than intended) perceptions of equipoise.

Supporting Information

S1 Data. Interview extracts illustrating evidence of 13 clinicians' less balanced views. (DOCX)

S1 Table. Consolidated Criteria for Reporting Qualitative Research (COREQ) completed checklist. (DOCX) S1 Text. Details of ethical approval.(DOCX)S2 Text. COREQ checklist additional information.

(DOCX)

Acknowledgments

Collaborating members of the RCT study groups are shown below.

ACST-2 study group: Alison Halliday (chief investigator), Jean-Pierre Becquemin, Anna Belli, Marc Bosiers, Piergiorgio Cao, Christina Davies, Jenny L. Donovan, Alastair Gray, Michael Gough, Elizabeth Hayter, Peter Leopold, Sumaira McDonald, Jonathan Michaels, Borislava Mihaylova, Richard Peto, Steven Robertson, Peter Rothwell, Rachael Scott, Dafydd Thomas, Frank Vermassen.

By-Band-Sleeve study group: Chris A. Rogers, Robert Andrews, Jane M. Blazeby (chief investigator), James Byrne, Jenny L. Donovan, Jamie Kelly, Graziella Mazza, David Mahon, Hamish Noble, Barnaby C. Reeves, Janice L. Thompson, Sarah Wordsworth, Richard Welbourn.

Chemorad study group: Jane M. Blazeby (chief investigator), Paul Barham, Sara T. Brookes, Tom Crosby, Jenny L. Donovan, Stephen J. Falk, S. Michael Griffin, William Holling-worth, Andrew D. Hollowood, Richard Krysztopik, Wyn Lewis, Jo Nicklin, Christopher Streets, Sean Strong, Dan Titcomb, Geraint Williams.

CSAW study group: David Beard (co-chief investigator), Andrew Carr (co-chief investigator), Jonathan Cook, Cushla Cooper, Benjamin Dean, Jenny L. Donovan, Alastair Gray, Stephen Gwilym, Andrew Judge, Naomi Merritt, Jane Moser, Jonathan Rees, Ines Rombach, Julian Savulescu, Irene Tracey, Karolina Wartolowska.

OPTIMA prelim study group: Rob Stein (chief investigator), John Bartlett, David Cameron, Amy Campbell, Peter Canney, Jenny L. Donovan, Janet Dunn, Helena Earl, Mary Falzon, Adele Francis, Peter Hall, Victoria Harmer, Helen Higgins, Louise Hiller, Luke Hughes-Davies, Claire Hulme, Iain Macpherson, Andreas Makris, Andrea Marshall, Christopher McCabe, Adrienne Morgan, Sarah Pinder, Christopher Poole, Elena Provenzano, Daniel Rea, Nigel Stallard.

POUT study group: Alison Birtle (chief investigator), Rik Bryan, James Catto, John Chester, Jenny L. Donovan, Ann French, Emma Hall, Chris Harris, Mark Johnson, Rob Jones, Francis Keeley, Tony Kirkbank, Roger Kockelbergh, Rebecca Lewis, Michelle Newton, Thomas Powles, Rachel Waters, Andrew Winterbottom.

Considerable thanks are given to the patient and clinician participants who agreed to be interviewed and audio-recorded in each RCT. We would also like to thank the research nurse teams who facilitated the recording and interview processes. The following individuals are acknowledged for their support of this work: Kerry Avery, Rebecca Barnes, Carmel Conefrey, Samantha Husbands, Nicola Mills, Katrina Turner, Ceri Rowlands, and Paul Whybrow.

The views expressed are those of the authors and not necessarily those of the Medical Research Council, Arthritis Research UK, Cancer Research UK, the National Health Service, the National Institute for Health Research, or the Department of Health.

Author Contributions

Conceptualization: JLD.

Data curation: LR.

Formal analysis: LR DE JW JLD.

Funding acquisition: DB JMB AB AH CAR RS JLD.

Investigation: LR DE MJ SP SS CW DB JMB AB AH CAR RS JLD.

Methodology: JLD.

Project administration: DB JMB AB AH CAR RS JLD.

Supervision: JLD.

Validation: LR DE JW JLD.

Writing - original draft: LR JLD.

Writing - review & editing: LR DE JW MJ SP SS CW DB JMB AB AH CAR RS JLD.

References

- Toerien M, Brookes ST, Metcalfe C, De Salis I, Tomlin Z, Peters TJ, et al. A review of reporting of participant recruitment and retention in RCTs in six major journals. Trials. 2009; 10:52. doi: 10.1186/1745-6215-10-52 PMID: 19591685
- Kasenda B, Von Elm E, You J, Blümle A, Tomonaga Y, Saccilotto R, et al. Prevalence, characteristics, and publication of discontinued randomized trials. JAMA. 2014; 311:1045–1052. doi: 10.1001/jama. 2014.1361 PMID: 24618966
- Kitterman DR, Cheng SK, Dilts DM, Orwoll ES. The prevalence and economic impact of low-enrolling clinical studies at an academic medical center. Acad Med. 2011; 86:1360. doi: 10.1097/ACM. 0b013e3182306440 PMID: 21952064
- Donovan JL, de Salis I, Toerien M, Paramasivan S, Hamdy FC, Blazeby JM. The intellectual challenges and emotional consequences of equipoise contributed to the fragility of recruitment in six randomized controlled trials. J Clin Epidemiol. 2014; 67:912–920. doi: 10.1016/j.jclinepi.2014.03.010
 PMID: 24811157
- Chard J, Lilford R. The use of equipoise in clinical trials. Soc Sci Med. 1998; 47:891–898. doi: 10.1016/ s0277-9536(98)00153-1 PMID: 9722109
- 6. Djulbegovic B. The paradox of equipoise: the principle that drives and limits therapeutic discoveries in clinical research. Cancer Control. 2009; 16:342–347. PMID: 19910921
- Horton R. Surgical research or comic opera: questions, but few answers. Lancet. 1996; 347:984–985. doi: 10.5555/uri:pii:S0140673696901373 PMID: 8606606
- Freedman B. Equipoise and the ethics of clinical research. N Engl J Med. 1987; 317:141–145. doi: 10. 1056/nejm198707163170304 PMID: 3600702
- 9. Miller FG, Brody H. A critique of clinical equipoise. Therapeutic misconception in the ethics of clinical trials. Hastings Cent Rep. 2003; 33:19–28. doi: 10.2307/3528434 PMID: 12854452
- Djulbegovic B, Lacevic M, Cantor A, Fields KK, Bennett CL, Adams JR, et al. The uncertainty principle and industry-sponsored research. Lancet. 2000; 356:635–638. doi: 10.1016/s0140-6736(00)02605-2 PMID: 10968436
- 11. Madsen S, Holm S, Riis P. Attitudes towards clinical research among cancer trial participants and nonparticipants: an interview study using a Grounded Theory approach. J Med Ethics. 2007; 33:234–240. doi: 10.1136/jme.2005.015255 PMID: 17400624
- 12. Jackson CJ, Dixon-Woods M, Eborall H, Kenyon S, Tooz-Hobson P, Tincello D. Women's views and experiences of a patient preference trial in surgery: a qualitative study of the CARPET1 trial. Clin Trials. 2010; 7:696–704. doi: 10.1177/1740774510381286 PMID: 20729253
- Mills N, Donovan JL, Smith M, Jacoby A, Neal DE, Hamdy FC. Perceptions of equipoise are crucial to trial participation: a qualitative study of men in the ProtecT study. Control Clin Trials. 2003; 24:272– 282. doi: 10.1016/s0197-2456(03)00020-5 PMID: 12757993
- Harrop E, Kelly J, Griffiths G, Casbard A, Nelson A. Why do patients decline surgical trials? Findings from a qualitative interview study embedded in the Cancer Research UK BOLERO trial (Bladder cancer: Open versus Lapararoscopic or RObotic cystectomy). Trials. 2016; 17:35. doi: <u>10.1186/s13063-</u> 016-1173-z PMID: 26787177

- Taylor KM. Integrating conflicting professional roles: physician participation in randomized clinical trials. Soc Sci Med. 1992; 35:217–224. doi: 10.1016/0277-9536(92)90169-q PMID: 1509310
- Lawton J, Kirkham J, White D, Rankin D, Cooper C, Heller S. Uncovering the emotional aspects of working on a clinical trial: a qualitative study of the experiences and views of staff involved in a type 1 diabetes trial. Trials. 2015; 16:3. doi: 10.1186/1745-6215-16-3 PMID: 25566971
- Potter S, Mills N, Cawthorn SJ, Donovan J, Blazeby JM. Time to be BRAVE: is educating surgeons the key to unlocking the potential of randomised clinical trials in surgery? A qualitative study. Trials. 2014; 15:80. doi: 10.1186/1745-6215-15-80 PMID: 24628821
- Eborall HC, Dallosso HM, Daly H, Martin-Stacey L, Heller SR. The face of equipoise—delivering a structured education programme within a randomized controlled trial: qualitative study. Trials. 2014; 15:6215–6215. doi: 10.1186/1745-6215-15-15 PMID: 24405854
- Miller FG, Joffe S. Equipoise and the dilemma of randomized clinical trials. N Engl J Med. 2011; 364:476. doi: 10.1056/NEJMsb1011301 PMID: 21288100
- Weijer C, Shapiro SH, Glass KC, Enkin MW. Clinical equipoise and not the uncertainty principle is the moral underpinning of the randomised controlled trial. BMJ. 2000; 321:756–759. PMID: 10999914
- 21. University of Bristol School of Social and Community Medicine. Qualitative Research Integrated within Trials (QuinteT). 2016 [cited 8 Jun 2016]. Available: <u>http://www.bristol.ac.uk/social-community-medicine/research/groups/social-sciences-health/quintet/</u>.
- Donovan JL, Rooshenas L, Jepson M, Elliott D, Wade J, Avery K, et al. Optimising recruitment and informed consent in randomised controlled trials: the development and implementation of the Quintet Recruitment Intervention (QRI). Trials. 2016; 17:1–11. doi: <u>10.1186/s13063-016-1391-4</u> PMID: 27278130
- Rogers CA, Welbourn R, Byrne J, Donovan JL, Reeves BC, Wordsworth S, et al. The By-Band study: gastric bypass or adjustable gastric band surgery to treat morbid obesity: study protocol for a multicentre randomised controlled trial with an internal pilot phase. Trials. 2014; 15:53. doi: 10.1186/1745-6215-15-53 PMID: 24517309
- Birtle AJ, Lewis R, Johnson M, Hall E. Time to define an international standard of postoperative care for resected upper urinary tract transitional cell carcinoma (TCC)—opening of the peri-operative chemotherapy versus surveillance in upper tract urothelial cancer (POUT) Trial. BJU Int. 2012; 110:919– 921. doi: 10.1111/j.1464-410X.2012.11413.x PMID: 22882350
- Bartlett J, Canney P, Campbell A, Cameron D, Donovan J, Dunn J, et al. Selecting breast cancer patients for chemotherapy: the opening of the UK OPTIMA trial. Clin Oncol. 2013; 25:109–116. doi: 10. 1016/j.clon.2012.10.005 PMID: 23267818
- Rudarakanchana N, Dialynas M, Halliday A. Asymptomatic Carotid Surgery Trial-2 (ACST-2): rationale for a randomised clinical trial comparing carotid endarterectomy with carotid artery stenting in patients with asymptomatic carotid artery stenosis. Eur J Vasc Endovasc Surg. 2009; 38:239–242. doi: 10. 1016/j.ejvs.2009.05.010 PMID: 19540137
- 27. Beard D, Rees J, Rombach I, Cooper C, Cook J, Merritt N, et al. The CSAW Study (Can Shoulder Arthroscopy Work?)—a placebo-controlled surgical intervention trial assessing the clinical and cost effectiveness of arthroscopic subacromial decompression for shoulder pain: study protocol for a randomised controlled trial. Trials. 2015; 16:210. doi: 10.1186/s13063-015-0725-y PMID: 25956385
- Blazeby J, Strong S, Donovan J, Wilson C, Hollingworth W, Crosby T, et al. Feasibility RCT of definitive chemoradiotherapy or chemotherapy and surgery for oesophageal squamous cell cancer. Br J Cancer. 2014; 111:234–240. doi: 10.1038/bjc.2014.313 PMID: 24921919
- Glaser BG, Strauss AL. The discovery of grounded theory: strategies for qualitative research. Chicago: Aldine; 1967.
- Donovan J, Mills N, Smith M, Brindle L, Jacoby A, Peters T, et al. Quality improvement report: improving design and conduct of randomised trials by embedding them in qualitative research: ProtecT (prostate testing for cancer and treatment) study. BMJ. 2002:766–769. PMID: 12364308
- Hamilton D, De Salis I, Donovan J, Birchall M. The recruitment of patients to trials in head and neck cancer: a qualitative study of the EaStER trial of treatments for early laryngeal cancer. Eur Arch Otorhinolaryngol. 2013; 270:2333–2337. doi: 10.1007/s00405-013-2349-8 PMID: 23334205
- Jenkins V, Fallowfield L, Souhami A, Sawtell M. How do doctors explain randomised clinical trials to their patients? Eur J Cancer. 1999; 35:1187–1193. doi: 10.1016/s0959-8049(99)00116-1 PMID: 10615228
- Byrne-Davis LM, Salmon P, Gravenhorst K, Eden TO, Young B. Balancing high accrual and ethical recruitment in paediatric oncology: a qualitative study of the'look and feel'of clinical trial discussions. BMC Med Res Methodol. 2010; 10:101. doi: 10.1186/1471-2288-10-101 PMID: 20969763

- Brown R, Butow P, Boyle F, Tattersall M. Seeking informed consent to cancer clinical trials; evaluating the efficacy of doctor communication skills training. Psychooncology. 2007; 16:507–516. PMID: 16986176
- Garcia J, Elbourne D, Snowdon C. Equipoise: a case study of the views of clinicians involved in two neonatal trials. Clin Trials. 2004; 1:170–178. doi: 10.1191/1740774504cn020xx PMID: 16281889
- Han PK, Klein WM, Arora NK. Varieties of uncertainty in health care a conceptual taxonomy. Med Decis Making. 2011; 31:828–838. doi: 10.1177/0272989X11393976 PMID: 22067431
- Griffiths F, Green E, Tsouroufli M. The nature of medical evidence and its inherent uncertainty for the clinical consultation: qualitative study. BMJ. 2005; 330:511. doi: <u>10.1136/bmj.38336.482720.8F</u> PMID: <u>15684026</u>
- Pickles K, Carter SM, Rychetnik L, McCaffery K, Entwistle VA. General practitioners' experiences of, and responses to, uncertainty in prostate cancer screening: insights from a qualitative study. PLoS ONE. 2016; 11:e0153299. doi: 10.1371/journal.pone.0153299 PMID: 27100402
- Tarn DM, Paterniti DA, Wenger NS. Provider recommendations in the face of scientific uncertainty: an analysis of audio-recorded discussions about vitamin D. J Gen Intern Med. 2016; 31:909–917. doi: 10. 1007/s11606-016-3667-5 PMID: 27008650
- 40. Elwyn G, Edwards A, Kinnersley P, Grol R. Shared decision making and the concept of equipoise: the competences of involving patients in healthcare choices. Br J Gen Pract. 2000; 50:892–899. PMID: 11141876
- Diamond-Brown L. The doctor-patient relationship as a toolkit for uncertain clinical decisions. Soc Sci Med. 2016; 159:108–115. doi: 10.1016/j.socscimed.2016.05.002 PMID: 27179146
- Donovan JL, Paramasivan S, de Salis I, Toerien M. Clear obstacles and hidden challenges: understanding recruiter perspectives in six pragmatic randomised controlled trials. Trials. 2014; 15:5. doi: 10.1186/1745-6215-15-5 PMID: 24393291
- Kurzrock R, Stewart DJ. Equipoise abandoned? Randomization and clinical trials. Ann Oncol. 2013; 24:2471–2474. doi: 10.1093/annonc/mdt358 PMID: 24072520
- Jenkins V, Fallowfield L, Solis-Trapala I, Langridge C, Farewell V. Discussing randomised clinical trials of cancer therapy: evaluation of a Cancer Research UK training programme. BMJ. 2005; 330:400. doi: 10.1136/bmj.38366.562685.8F PMID: 15705666
- 45. Paramasivan S, Strong S, Wilson C, Campbell B, Blazeby JM, Donovan JL. A simple technique to identify key recruitment issues in randomised controlled trials: Q-QAT—Quanti-Qualitative Appointment Timing. Trials. 2015; 16:88. doi: 10.1186/s13063-015-0617-1 PMID: 25873096